Page1 CS248 Lecture 1Copyright � Pat HanrahanMotivationInteractive graphical user interfacesnMouse, windows and widgetsnVirtual realityInput: Gloves, 3D trackingOuput: HMD

Copyright � Pat HanrahanMotivationDigital media technologiesnPrintingnFilm and videon

Multimedia workstation

Page2 CS248 Lecture 1Copyright � Pat HanrahanMotivationVisualizationnScience and Engineering: finite element simulationnMedicine: CT ScansnMathematics: 4D geometry, minimal surfaces

Physics: General relativity n Chemistry: Macromolecules

CS248 Lecture 1Copyright � Pat HanrahanMotivationArt and EntertainmentnMovies: Juraisic Park and Toy StorynCDROM Titles

Page3 CS248 Lecture 1Copyright � Pat HanrahanInterdisciplinary: Theory and PracticeSciencenMathematics of curves and surfacesnPhysics of light and colorEngineeringn

n Software: Graphics libraries, window systems n Displays

Aethestics n Color theory n Composition ...

CS248 Lecture 1Copyright � Pat HanrahanGraphics Pipeline: 3D Geometric Graphics

Lighting and Shading

Hidden Surface Elimination

Animation: time

Page4 CS248 Lecture 1Copyright � Pat HanrahanGraphics Pipeline: 2D Raster Graphics

Texture Mapping

Imaging/Compositing

Intensity and ColorQuantization

Framebuffer

Scan Conversion